Author's response to reviews

Title: Dietary intake of Senegalese adults

Authors:

- Cheryl A. M. Anderson (chanders@jhsph.edu)
- Scarlette Bellamy (bellamys@mail.med.upenn.edu)
- Mindy Figures (Mindy2@mail.med.upenn.edu)
- Charnita Zeigler-Johnson (czi@mail.med.upenn.edu)
- Mohamed Jallow (jmohamed60@yahoo.fr)
- Elaine Spangler (espangle@mail.med.upenn.edu)
- Margerie Coomes (margerie@mail.med.upenn.edu)
- Serigne Gueye (smgueye@sentoo.sn)
- Timothy Rebbeck (rebbeck@mail.med.upenn.edu)

Version: 4 Date: 21 January 2010

Author's response to reviews: see over
Dear Editor,

I summarized the changes made to our manuscript titled “Dietary intake of Senegalese adults” (manuscript number 3072536129431010) below.

1. We updated each author’s affiliation so that it is written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. The authors’ names are listed at the top of the title page, with the affiliations listed underneath and linked to the appropriate authors via superscript numbers.

2. The list of authors in the manuscript is written exactly as in the submission system, both in style and order.

3. References are cited in the text using consecutive numbers in square brackets, and not superscript numbers. All references were placed before the full stop.

4. We included a year in the details for reference 7.

5. We included a 'Competing Interests' section between the Conclusions and Authors’ Contributions.

6. We deleted the following text from the title page: 'title page:' and 'title'.

7. We changed the sentence case in the title, both in the manuscript file and the submission system.

8. Keywords were deleted from the title page.

9. We removed the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system.

10. We deleted the repeated main title above the abstract section.

11. Visible vertical lines were removed from our tables.
12. We checked for typographical errors. No additional corrections (besides those listed above) were made during our check.

Sincerely,

Cheryl A. M. Anderson